Author's response to reviews

Title: Reported food intake and distribution of body fat: A repeated cross-sectional study

Authors:

Benno Krachler (benno.krachler@medicin.umu.se)
Mats Eliasson (Mats.Eliasson@nll.se)
Hans Stenlund (hans.stenlund@epiph.umu.se)
Ingegerd Johansson (Ingegerd.Johansson@odont.umu.se)
Goran Hallmans (goran.hallmans@nutires.umu.se)
Bernt Lindahl (bernt.lindahl@medicin.umu.se)

Version: 6 Date: 21 December 2006

Author's response to reviews:

A further, small change in table 1: Since we, as requested removed the SAS program procedure the second column of the table could be simplified. Thus it will fit more readily on one page.

Yours, Benno Krachler